\documentclass[10pt,a4paper]{amsart}
\usepackage{amsfonts,amsmath,amssymb}
\usepackage{hyperref}
\usepackage[all]{xy}
\newtheorem*{theorem*}{Theorem}
\newtheorem{lemma}{Lemma}[subsection]
\newtheorem{proposition}[lemma]{Proposition}
\newtheorem{remark}[lemma]{Remark}
\newtheorem{example}[lemma]{Example}
\newtheorem{theorem}[lemma]{Theorem}
\newtheorem{definition}[lemma]{Definition}
\newtheorem{notation}[lemma]{Notation}
%\newtheorem{result}{Result}[section]
\newtheorem{property}[lemma]{Property}
\newtheorem{corollary}[lemma]{Corollary}
\newtheorem{construction}[lemma]{Construction}
\newtheorem{case}{Case}
\newtheorem{conjecture}{Conjecture}
\newtheorem*{conjecture*}{Conjecture}
\oddsidemargin=0cm
%
\evensidemargin=0cm
%\textheight=22cm
\baselineskip 18pt \textwidth 16cm \sloppy \theoremstyle{plain}
\newcommand{\ram}{\operatorname{ram}}
\newcommand{\tr}{\operatorname{Tr}}
\newcommand{\www}{\mathfrak{W}}
\newcommand{\ob}{\operatorname{ ob}}
\newcommand{\ind}{\operatorname{ind}}
\newcommand{\bc}{\operatorname{bc}}
\newcommand{\symm}{\mathbb{S}}
\newcommand{\res}{\operatorname{Res}_{s=1}}
\newcommand{\ad}{\operatorname{ad}}
\newcommand{\bes}{\mathfrak{B}}
\newcommand{\eisen}{\mathcal{E}}
\newcommand{\rest}{\lvert}
\newcommand{\dd}{\mathfrak{D}}
\newcommand{\uu}{\mathcal{U}}
\newcommand{\hh}{\mathcal{H}}
\newcommand{\ff}{\mathcal{F}}
\newcommand{\jj}{\mathfrak{J}}
\newcommand{\pp}{\mathfrak{p}}
\newcommand{\oo}{\mathcal{O}}
\renewcommand{\P}{\mathcal{P}}
\newcommand{\ch}{{\bf 1}}
\newcommand{\sss}{\mathcal{S}}
\newcommand{\ww}{\mathcal{W}}
\newcommand{\bb}{\mathcal{B}}
\newcommand{\ii}{\mathcal{I}}
\newcommand{\LL}{\mathcal{L}}
\newcommand{\Ker}{\mathfrak{K}}
\newcommand{\ccc}{\mathcal{C}}
\newcommand{\nm}{\operatorname{Nm}}
\newcommand{\bs}{\backslash}
\newcommand{\swrtz}{\mathcal{S}}
\newcommand{\Hom}{\operatorname{Hom}}
\newcommand{\diag}{\operatorname{diag}}
\newcommand{\sgn}{\operatorname{sgn}}
\newcommand{\kk}{{\bf K}}
\newcommand{\zz}{\mathbb{Z}}
\newcommand{\nn}{\mathbb{N}}
\newcommand{\cc}{\mathbb{C}}
\newcommand{\qq}{\mathbb{Q}}
\newcommand{\rr}{\mathbb{R}}
\newcommand{\disc}{\Delta}
\newcommand{\modulus}{\delta}
\newcommand{\aaa}{\mathfrak{a}}
\newcommand{\aut}{\mathcal{A}}
\newcommand{\A}{\mathbb{A}}
\newcommand{\I}{\mathbb{I}}
\newcommand{\eps}{\varepsilon}
\newcommand{\vol}{\operatorname{vol}}
\newcommand{\Ht}{H}
\newcommand{\schwartzp}{C_c^\infty(M_n(F))}
\newcommand{\schwartz}{C_c^\infty(H_n(E/F))}
\newcommand{\diff}{\mathfrak{d}}
\newcommand{\triv}{{\bf 1}}
\newcommand{\stab}{\operatorname{Stab}}
\newcommand{\Ind}{\operatorname{Ind}}
\newcommand{\id}{\operatorname{Id}}
\newcommand{\re}{\operatorname{Re}}
\renewcommand{\Im}{\operatorname{Im}}
\newcommand{\vrel}{{v_{\operatorname{rel}}}}
\newcommand{\vaux}{{v_{\operatorname{aux}}}}
\newcommand{\vcusp}{{v_{\operatorname{cusp}}}}
\newcommand{\match}[1]{\mathop{\longleftrightarrow}\limits^{#1}}
\newcommand{\pperp}{\hbox{$\perp\hskip-6pt\perp$}}
\newcommand{\N}{{\mathbb N}}
\newcommand{\PP}{{\mathbb P}}
\newcommand{\Z}{{\mathbb Z}}
\newcommand{\Q}{{\mathbb Q}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\C}{{\mathbb C}}
\newcommand{\K}{{\mathbb K}}
\newcommand{\F}{{\mathbb F}}
\newcommand{\E}{{\mathcal E}}
\newcommand{\proofend}{\hfill$\Box$\smallskip}
\newcommand{\ko}{{\mathcal O}}
\newcommand{\wx}{{\widetilde x}}
\newcommand{\wz}{{\widetilde z}}
\newcommand{\wa}{{\widetilde a}}
\newcommand{\bz}{{\boldsymbol z}}
\newcommand{\bp}{{\boldsymbol p}}
\newcommand{\wy}{{\widetilde y}}
\newcommand{\wc}{{\widetilde c}}
\newcommand{\bi}{{\omega}}
\newcommand{\bx}{{\boldsymbol x}}
\newcommand{\Log}{{\operatorname{Log}}}
\newcommand{\pr}{{\operatorname{pr}}}
\newcommand{\Graph}{{\operatorname{Graph}}}
\newcommand{\jet}{{\operatorname{jet}}}
\newcommand{\Tor}{{\operatorname{Tor}}}
\newcommand{\sqh}{{\operatorname{sqh}}}
\newcommand{\const}{{\operatorname{const}}}
\newcommand{\Arc}{{\operatorname{Arc}}}
\newcommand{\Sing}{{\operatorname{Sing}}}
\newcommand{\Span}{{\operatorname{Span}}}
\newcommand{\Aut}{{\operatorname{Aut}}}
\newcommand{\Int}{{\operatorname{Int}}}
\newcommand{\Aff}{{\operatorname{Aff}}}
\newcommand{\Area}{{\operatorname{Area}}}
\newcommand{\val}{{\operatorname{Val}}}
\newcommand{\conv}{{\operatorname{conv}}}
\newcommand{\rk}{{\operatorname{rk}}}
\newcommand{\End}{\operatorname{End}}
\newcommand{\ow}{{\overline w}}
\newcommand{\ov}{{\overline v}}
%\newcommand{\Sc}{{\mathcal S}}
\newcommand{\G}{{\mathcal G}}
\newcommand{\T}{{\mathcal T}}
\newcommand{\red}{{\operatorname{red}}}
\newcommand{\kc}{{\mathcal C}}
\newcommand{\ki}{{\mathcal I}}
\newcommand{\kj}{{\mathcal J}}
\newcommand{\ke}{{\mathcal E}}
\newcommand{\kz}{{\mathcal Z}}
\newcommand{\tet}{{\theta}}
\newcommand{\Del}{{\Delta}}
\newcommand{\bet}{{\beta}}
\newcommand{\mm}{{\mathfrak m}}
\newcommand{\kap}{{\kappa}}
\newcommand{\del}{{\delta}}
\newcommand{\sig}{{\sigma}}
\newcommand{\alp}{{\alpha}}
\newcommand{\Sig}{{\Sigma}}
\newcommand{\Gam}{{\Gamma}}
\newcommand{\gam}{{\gamma}}
\newcommand{\Lam}{{\Lambda}}
\newcommand{\lam}{{\lambda}}
\newcommand{\om}{{\omega}}
\newcommand{\Fre}{{Fr\'{e}chet \,}}
\newcommand{\et}{{\'{e}tale }}
\newcommand{\tG}{{\widetilde{G}}}
\newcommand{\tH}{{\widetilde{H}}}
\newcommand{\tO}{{\widetilde{O}}}
\newcommand{\Fou}{{\mathcal{F}}}
\newcommand{\abs}[1]{\left|{#1}\right|}
\newcommand{\norm}[1]{\lVert#1\rVert}
\newcommand{\sprod}[2]{\left\langle#1,#2\right\rangle}
\newcommand{\cC}{{C_c^{\infty}}}
\newcommand{\cD}{{\mathcal{D}}}
\newcommand{\g}{{\mathfrak{g}}}
\newcommand{\h}{{\mathfrak{h}}}
\newcommand{\z}{{\mathfrak{z}}}
\newcommand{\cgot}{{\mathfrak{c}}}
\newcommand{\Supp}{\mathrm{Supp}}
\newcommand{\uG}{{\underline{G}}}
\newcommand{\uH}{{\underline{H}}}
\newcommand{\uX}{{\underline{X}}}
\newcommand{\uU}{{\underline{U}}}
\newcommand{\uV}{{\underline{V}}}
\newcommand{\uW}{{\underline{W}}}
\newcommand{\uY}{{\underline{Y}}}
\newcommand{\uZ}{{\underline{Z}}}
\newcommand{\gd}{\g^{\sigma}}
\newcommand{\oF}{{\overline{F}}}
\newcommand{\cO}{{\mathcal{O}}}
\newcommand{\GL}{\operatorname{GL}}
\newcommand{\SL}{\operatorname{SL}}
\newcommand{\gl}{{\mathfrak{gl}}}
\newcommand{\sll}{{\mathfrak{sl}}}
\newcommand{\HC}{\operatorname{HC}}
\newcommand{\Fr}{\operatorname{Fr}}
\newcommand{\Mp}{\operatorname{Mp}}
%\newcommand{\Aut}{\operatorname{Aut}}
\newcommand{\Sym}{\operatorname{Sym}}
\newcommand{\Spec}{\operatorname{Spec}}
\newcommand{\Sc}{{\mathcal S}}
\newcommand{\Lie}{\operatorname{Lie}}
%\newcommand{\ad}{\operatorname{ad}}
\newcommand{\Ad}{\operatorname{Ad}}
\usepackage[usenames]{color}
%\newcommand{\Rami}[1]{{\color{red}{#1}}}
%\newcommand{\Dima}[1]{{\color{magenta}{#1}}}

\renewcommand{\cc}{\mathbb{C}}
\renewcommand{\Fre}{{Fr\'{e}chet \,}}
\renewcommand{\R}{{\mathbb R}}
\renewcommand{\cD}{{\mathcal{D}}}
\renewcommand{\Sc}{{\mathcal S}}
\renewcommand{\G}{{\mathcal G}}
\renewcommand{\T}{{\mathcal T}}

\newtheorem{result}{Result}%[subsection]
\newtheorem{property2}{}
%\newtheorem{remark}{Remark}%[subsection]{Remark}

\begin{document}
\begin{thebibliography}{99}
%\bibitem[BCR]{BCR} Bochnak, J; Coste, M; Roy, M-F:
%{\it Real Algebraic Geometry} (1998)
\bibitem[Cas1]{Cas} Casselman, W: {\it Introduction to the Schwartz Space of $\Gamma \setminus G$}. Canadian
Journal Mathematics Vol.\textbf{XL}, No2 , 285-320 (1989)
%\bibitem[Cas2]{Cas2} Casselman,W: {\it Bruhat Filtration}.
%Unpublished
\bibitem[CHM]{CHM} Casselman, William; Hecht, Henryk; Mili\v ci\'c, Dragan: \\
{\it Bruhat filtrations and Whittaker vectors for real groups}.
The mathematical legacy of Harish-Chandra (Baltimore, MD, 1998),
151-190, Proc. Sympos. Pure Math., \textbf{68}, Amer. Math. Soc.,
 Providence, RI, (2000)
\bibitem[dCl]{dCl} du Cloux, Fokko: {\it Sur les repr�sentations diff�rentiables des groupes de Lie
alg�briques}. Annales scientifiques de l��.N.S. 4e s�rie, tome
\textbf{24}, no 3, p. 257-318 (1991).
%\bibitem[DK]{DK} Delfs, H; Knebush, M: {\it Semialgebraic Topology Over a Real Closed Field II.} Mathematische Zeitschrift
%\textbf{178}, 175-213 (1981)
%%\bibitem[Dlz]{Dlz} Delzell,C.N.: {\it A Finiteness Theorem for Open Semi-algebraic Sets With Applications to Hilbert
%%s $17^{th}$ Problem} Contemporary Mathematics \textbf{8}, 79-97
%%(1982)
%\bibitem[HC]{HC} Harish-Chandra: {\it Automorphic Forms on Semi-Simple Lie Groups}. Bulletin of American Mathematical Society \textbf{76}, 529-551 (1970)
\bibitem[KS]{KS} Kashiwara, M.; Schapira, P.: {\it Ind-Sheaves}. Asterisques \textbf{271}, (2001)
%\bibitem[Mal]{Mal} Malgrange, B: {\it Ideals of Differentiable Functions}, Oxford University Press (1966)
\bibitem[Mor]{Mor} Morando, G.:{\it An existence theorem for tempered
solutions of D-modules on complex curves}, arXiv: math.AG/0505498
%(2006)
\bibitem[Pre]{Pre} Prelli, L.: {\it Sheaves on subanalytic sites}, arXiv: math.AG/0505498 (2007)
%\bibitem[Shi]{Shi} Shiota, M: {\it Nash Manifolds}.  Lecture Notes in Mathematics \textbf{1269} (1987)
\bibitem[vD]{vD} van Dijk {\it On a class of generalized Gelfand
pairs} Math. Z. 193, 581-593 (1986).
\end{thebibliography}


\author{Avraham Aizenbud}
%\address{Avraham Aizenbud, Faculty of Mathematics
%and Computer Science, The Weizmann Institute of Science POB 26,
%Rehovot 76100, ISRAEL.} 

%\email{aizenr@yahoo.com}
\title[Schwartz functions on Nash manifolds]{Schwartz functions on Nash
manifolds and applications to representation theory}
\maketitle
\centerline{Joint with Dmitry Gourevitch and Eitan Sayag} 
\centerline{arXiv:0704.2891 [math.AG], arxiv:0709.1273 [math.RT]}

$ $\\

Let us start with the following motivating example. Consider the
circle $S^1$, let $N \subset S^1$ be the north pole and denote
$U:=S^1 \setminus N$. Note that $U$ is diffeomorphic to $\R$ via
the stereographic projection. Consider the space $\cD(S^1)$ of
distributions on $S^1$, that is the space of continuous linear
functionals on the \Fre space $C^{\infty}(S^1)$. Consider the
subspace $\cD_{S^1}(N) \subset \cD(S^1)$ consisting of all
distributions supported at $N$. Then the quotient $ \cD(S^1) /
\cD_{S^1}(N)$ will not be the space of distributions on $U$.
However, it will be the space $\Sc^*(U)$ of Schwartz distributions
on $U$, that is continuous functionals on the \Fre space $\Sc(U)$
of Schwartz functions on $U$. In this case, $\Sc(U)$ can be
identified with $\Sc(\R)$ via the stereographic projection.

The space of Schwartz functions on $\R$ is defined to be the space
of all infinitely differentiable functions that rapidly decay at
infinity together with all their derivatives, i.e. $x^nf^{(k)}$ is
bounded for any $n,k$.

In this talk we extend the notions of Schwartz functions and
Schwartz distributions to a larger geometric realm.

%\subsection{}
As we can see, the definition is of algebraic nature. Hence it
would not be reasonable to try to extend it to arbitrary smooth
manifolds. However, it is reasonable to extend this notion to
smooth algebraic varieties. Unfortunately, sometimes this is not
enough. For example, a connected component of real algebraic
variety is not always an algebraic variety. By this reason we
extend this notion to smooth semi-algebraic manifolds. They are
called \textbf{Nash manifolds}.

For any Nash manifold $M$, we will define the spaces $\G(M)$,
$\T(M)$ and $\Sc(M)$ of \textbf{generalized Schwartz
functions}\footnote{Here we distinguish between the (similar)
notions of a generalized function and a distribution. They can be
identified by choosing a measure. Without fixing a measure, a
smooth function defines a generalized function but not a
distribution.}, \textbf{tempered functions} and \textbf{Schwartz
functions} on $M$. Informally, $\T(M)$ is the ring of functions
that have no more than polynomial growth together with all their
derivatives, $\G(M)$ is the space of generalized functions with no
more than polynomial growth and $\Sc(M)$ is the space of functions
that decay together with all their derivatives faster than any
inverse power of a polynomial.

As in the classical case, in order to define generalized Schwartz
functions, we have to define Schwartz functions first. Both
$\G(M)$ and $\Sc(M)$ are modules over $\T(M)$.

The triple $\Sc(M)$, $\T(M)$, $\G(M)$ is analogous to
$C_c^{\infty}(M)$, $C^{\infty}(M)$ and $C^{-\infty}(M)$ but it has
additional nice properties as we will see later.

We will show that for $M=\R^n$, $\Sc(M)$ is the space of classical
Schwartz functions and $\G(M)$ is the space of classical
generalized Schwartz functions. For compact Nash manifold $M$,
$\Sc(M)=\T(M)=C^{\infty}(M)$.

\subsection*{Main results}

%In this subsection we summarize the main results of the paper.
\begin{result}
Let $M$ be a Nash manifold and $Z\subset M$ be a closed Nash
submanifold. Then the restriction maps $\T(M) \rightarrow \T(Z)$
and $\Sc(M) \rightarrow \Sc(Z)$ are onto.
\end{result}

\begin{result} \label{mainresult}
Let $M$ be a Nash manifold and $U \subset M$  be a semi-algebraic
open subset. Then a Schwartz function on $U$ is the same as a
Schwartz function on $M$ which vanishes with all its derivatives
on $M \setminus U$.
\end{result}

This theorem tells us that extension by zero $\Sc(U)\rightarrow
\Sc(M)$ is a closed imbedding, and hence restriction morphism
$\G(M)\rightarrow \G(U)$ is onto.
\\Classical generalized functions do not have this property. This was our
main motivation for extending the definition of Schwartz
functions.

\subsection*{Schwartz sections of Nash bundles}

Similar notions will be defined for Nash bundles, i.e. smooth
semi-algebraic bundles.

For any Nash bundle $E$ over $M$ we will define the spaces
$\G(M,E)$, $\T(M,E)$ and $\Sc(M,E)$ of generalized Schwartz,
tempered and Schwartz sections of $E$.

As in the classical case, a generalized Schwartz function is not
exactly a functional on the space of Schwartz functions, but a
functional on Schwartz densities, i.e. Schwartz sections of the
bundle of densities.

Therefore, we will define generalized Schwartz sections by
$\G(M,E)=(\Sc(M,\widetilde {E}))^*$, where $\widetilde
{E}=E^*\otimes D_M$ and $D_M$ is the bundle of densities on M.

Let $Z \subset M$ be a closed Nash submanifold, and $U=M \setminus
Z$. Result \ref{mainresult} tells us that the quotient space of
$\G(M)$ by the subspace $\G(M)_Z$ of generalized Schwartz
functions supported in $Z$ is $\G(U)$. Hence it is useful to study
the space $\G(M)_Z$. As in the classical case, $\G(M)_Z$ has a
filtration by the degree of transversal derivatives of delta
functions. The quotients of the filtration are generalized
Schwartz sections over $Z$ of symmetric powers of normal bundle to
$Z$ in $M$, after a twist.

This result can be extended to generalized Schwartz sections of
arbitrary Nash bundles.

\subsection*{Restricted topology and sheaf properties}

Similarly to algebraic geometry, the reasonable topology on Nash
manifolds to consider is a topology in which open sets are open
semi-algebraic sets. Unfortunately, it is not a topology in the
usual sense of the word, it is only what is called
\textbf{restricted topology}. This means that the union of an
infinite number of open sets does not have to be open. The only
open covers considered in the restricted topology are finite open
covers.

The restriction of a generalized Schwartz function (respectively
of a tempered function) to an open subset is again a generalized
Schwartz (respectively a tempered function). This means that they
form pre-sheaves. We will show that they are actually sheaves,
which means that for any finite open cover $M = \bigcup \limits
_{i=1}^n U_i$, a function $\alpha$ on $M$ is tempered if and only
if $\alpha|_{U_i}$ is tempered for all $i$. It is of course not
true for infinite covers. We denote the sheaf of generalized
Schwartz functions by $\G_M$ and of the sheaf of tempered
functions by $\T_M$. By result \ref{mainresult}, $\G_M$ is a
flabby sheaf.

Similarly, for any Nash bundle $E$ over $M$ we will define the
sheaf $\T_M^E $ of tempered sections and the sheaf $\G_M^E$ of
generalized Schwartz sections.

As we have mentioned before, Schwartz functions behave similarly
to compactly supported smooth functions. In particular, they
cannot be restricted to an open subset, but can be extended by
zero from an open subset. This means that they do not form a
sheaf, but an object dual to a sheaf, a so-called cosheaf. We
denote the cosheaf of Schwartz functions by $\Sc_M$.
%we will show that if $U$ is an open (semi-algebraic) subset of
%$M$, the extension by zero of a Schwartz function on $U$ is a
%Schwartz function on $M$.
%This property shows us the structure of
%cosheaf on Schwartz functions on $M$. Cosheaf is a notion dual to
%the notion of sheaf, i.e. instead of restriction to open subset we
%have extension from open subset.
We will prove that $\Sc_M$ is actually a cosheaf and not just
pre-cosheaf by proving a Schwartz version of the partition of
unity theorem. Similarly, for any Nash bundle $E$ over $M$ we will
define the cosheaf $\Sc_M^E$ of Schwartz sections.

\subsection*{Possible applications}

Schwartz functions are used in the representation theory of
algebraic groups. Our definition coincides with Casselman's
definition (cf. \cite{Cas}) for algebraic groups. Our paper allows
to use Schwartz functions in more situations in the representation
theory of algebraic groups, since an orbit of an algebraic action
is a Nash manifold, but does not have to be an algebraic group or
even an algebraic variety.

Generalized Schwartz sections can be used for ``devisage''. We
mean the following. Let $U\subset M$ be an open (semi-algebraic)
subset. Instead of dealing with generalized Schwartz sections of a
bundle on $M$, we can deal with generalized Schwartz sections of
its restriction to $U$ and generalized Schwartz sections of some
other bundles on $M \setminus U$.

For example if we are given an action of an algebraic group $G$ on
an algebraic variety $M$, and a $G$-equivariant bundle $E$ over
$M$, then devisage to orbits helps us to investigate the space of
$G$-invariant generalized sections of $E$.

%One of the implementations of this argument appears in \cite{AGS}.
%There we also use the fact that $\Sc(\R^n)$ is preserved by
%Fourier transform.

\subsection*{Summary} \label{Summary}

To sum up, for any Nash manifold $M$ we define a sheaf $\T_M$ of
algebras on $M$ (in the restricted topology) consisting of
tempered functions, a sheaf $\G_M$ of modules over $\T_M$
consisting of generalized Schwartz functions, and a cosheaf
$\Sc_M$ of modules over $\T_M$ consisting of Schwartz functions.

Moreover, for any Nash bundle $E$ over $M$ we define sheaves
$\T^E_M$ and $\G^E_M$ of modules over $\T_M$ consisting of
tempered and generalized Schwartz sections of $E$ respectively and
a cosheaf $\Sc^E_M$ of modules over $\T_M$ consisting of Schwartz
sections of $E$.

Let us list the main properties of these objects that we will
prove in this paper:

\begin{property2}\label{p1}
Compatibility: For open semi-algebraic subset $U \subset M$,
$\Sc^E_M|_U=\Sc^{E|_U}_U$, $\T^E_M|_U=\T^{E|_U}_U$,
$\G^E_M|_U=\G^{E|_U}_U$.
\end{property2}
%
\begin{property2}\label{p2}
$\Sc(\R ^n)$ = Classical Schwartz functions on $\R ^n$.
\end{property2}
%
\begin{property2}\label{p3}
For compact $M$, $\Sc(M,E) = \T(M,E)=$ $C^{\infty}(M,E)$.
\end{property2}
%
\begin{property2}\label{p4}
$\G_M^{E}=(\Sc^{\widetilde {E} }_M)^*$ , where $\widetilde
{E}=E^*\otimes D_M$ and $D_M$ is the bundle of densities on M.
\end{property2}
%
\begin{property2}\label{p5}
Let $Z \subset M$ be a closed Nash submanifold. Then the
restriction maps  $\Sc(M,E)$ onto $\Sc(Z,E|_Z)$ and $\T(M,E)$ onto
$\T(Z,E|_Z)$.
\end{property2}
%
\begin{property2}\label{p6}
Let $U \subset M$  be a semi-algebraic open subset, then
$$\Sc^E_M(U) \cong \{\phi \in \Sc(M,E)| \quad \phi \text{ is 0 on } M
\setminus U \text{ with all derivatives} \}.$$
\end{property2}
%
\begin{property2} \label{p7}
Let $Z \subset M$ be a closed Nash submanifold. Consider
$\G(M,E)_Z=\{\xi \in \G(M,E) |\xi$ is supported in $Z \}$. It has
a canonical filtration such that its factors are canonically
isomorphic to $\G(Z,{E|_Z \otimes S^i(N_Z^M) \otimes {D_M^*|}_Z
\otimes D_Z})$ where $N_Z^M$ is the normal bundle of $Z$ in $M$
and $S^i$ means i-th symmetric power.
\end{property2}
%
\subsection*{Remarks}
%
\begin{remark}
Harish-Chandra has defined a Schwartz space for every reductive
Lie group. However, Harish-Chandra's Schwartz space does not
coincide with the space of Schwartz functions that we define in
this paper even for the algebraic group $\R^{\times}$.

%Harish-Chandra gave a different definitions of Schwartz functions,
%generalized Schwartz functions and tempered functions on Lie
%groups (cf. \cite{HC}).
%
%These definitions are not equivalent to our definition, even for
%the algebraic group $\R^{\times}$.
%Our definition does not coincide with Harish-Chandra's definition
%(cf. \cite{HC}) for Lie groups, because a Lie group isomorphism
%which is not a Nash diffeomorphism maps Harish-Chandra's Schwartz
%functions to Harish-Chandra's Schwartz functions but will not map
%our Schwartz functions to our Schwartz functions.
\end{remark}
%
\begin{remark}
There is a different approach to the concept of Schwartz
functions. Namely, if $M$ is embedded as an open subset in a
compact manifold $K$ then one can define the space of Schwartz
functions on $M$ to be the space of all smooth functions on $K$
that vanish outside $M$ together with all their derivatives. This
approach is implemented in \cite{CHM}, \cite{KS}, \cite{Mor} and
\cite{Pre}. In general, this definition depends on the embedding
into $K$. Our results show that for Nash manifolds $M$ and $K$ it
coincides with our definition and hence does not depend on the
embedding.
\end{remark}
%

\begin{remark}
After the completion of this project we found out that many of the
properties of Schwartz functions on affine Nash manifolds have
been obtained already in \cite{dCl}.
\end{remark}

%\begin{remark}
%In fact, tempered functions can be defined in terms of Schwartz
%functions or in terms of generalized Schwartz functions by
%$$\T(M,E) = \{\alpha \in
%C^{\infty}(M,E)|\,\,\, \alpha \Sc(M)\subset \Sc(M,E)\} =$$
%$$\quad \quad \quad =\{\alpha \in C^{\infty}(M,E) |\,\,\, \alpha \G(M)\subset
%\G(M,E)\},$$ but the proof is rather technical and lies out of the
%scope of this paper.
%\end{remark}
%
%\begin{remark}
%Throughout the whole paper ``smooth'' means infinitely
%differentiable and ``bounded'' means bounded in absolute value,
%unless stated otherwise.
%\end{remark}
\subsection*{An application to representation theory}

Using the theory of Schwartz functions we showed the following
theorem:
\begin{theorem}
Let $F$ be either the field of real or complex numbers and
consider the standard imbedding $\mathrm{GL}_n(F) \hookrightarrow
\mathrm{GL}_{n+1}(F)$. We consider the two-sided action of
$\mathrm{GL}_n(F) \times \mathrm{GL}_n(F)$ on
$\mathrm{GL}_{n+1}(F)$ defined by
%$(g_1,g_2)h:=j(g_1)hj(g_2^{-1})$
$(g_1,g_2)h:=g_1 hg_2^{-1}$. Then any $\mathrm{GL}_n(F) \times
\mathrm{GL}_n(F)$ invariant distribution on $\mathrm{GL}_{n+1}(F)$
is invariant with respect to transposition.
\end{theorem}
This theorem has the following corollary in representation theory.

\begin{theorem}.
Let $(\pi,E)$ be an irreducible admissible continuous
representation of $\mathrm{GL}_{n+1}(F)$ on a Hilbert space $E$.
Then
\begin{equation}\label{dim1}
dim Hom_{\mathrm{GL}_n(F)}(E^{\infty},\cc) \leq 1
\end{equation}
\end{theorem}

Clearly, the last theorem implies in particular that (\ref{dim1})
holds for unitary irreducible representations of
$\mathrm{GL}_{n+1}(F)$. That is, the pair
$(\mathrm{GL}_{n+1}(F),\mathrm{GL}_{n}(F))$ is a generalized
Gelfand pair in the sense of \cite{vD}.



\end{document}
